Author's response to reviews

Title: Tono-Pen XL tonometry during application of a suction ring in rabbits

Authors:

Spyridon K Charisis (scharisis@gmail.com)
Harilaos S Ginis (ginis@ivo.gr)
Georgios A Kounis (gkounis@med.uoc.gr)
Miltiadis K Tsilimbaris (tsilimb@med.uoc.gr)

Version: 3 Date: 29 June 2008

Author's response to reviews:

Heraklion June 29, 2008

Dear Editor,

We would like to thank again the Editor and the Reviewer their comments. R2 wherever it appeared in the document was converted to Rsuperscript 2. We have also made a few corrections so that our revised manuscript conforms to the journal style. It would be helpful if you could provide us with a preformal invoice, if our article is accepted for publication.

With kind regards,

Spyridon K Charisis
Resident in Ophthalmology
University Hospital of Heraklion, Eye Clinic
Heraklion, Crete, Greece
scharisis@gmail.com